Author's response to reviews

Title: Lack of association of genetic variation in chromosome region 15q14-22.1 with type 2 diabetes in a Japanese population

Authors:

Yuka Yamaguchi (yama@genome.tokushima-u.ac.jp)
Maki Moritani (moritani@genome.tokushima-u.ac.jp)
Toshihito Tanahashi (tana@genome.tokushima-u.ac.jp)
Dai Osabe (d_osabe@ip.fujitsu.com)
Kyoko Nomura (nomura.kyoko@jp.fujitsu.com)
Yuka Fujita (fujita@genome.tokushima-u.ac.jp)
Parvaneh Keshavarz (parvaneh@genome.tokushima-u.ac.jp)
Kiyoshi Kunika (kunika@genome.tokushima-u.ac.jp)
Naoto Nakamura (naoto@koto.kpu-m.ac.jp)
Toshikazu Yoshikawa (toshi@koto.kpu-m.ac.jp)
Eiichiro Ichiishi (ichiishi@XG7.SO-NET.ne.jp)
Hiroshi Shiota (shiota@clin.med.tokushima-u.ac.jp)
Natsuo Yasui (nyasui@clin.med.tokushima-u.ac.jp)
Hiroshi Inoue (hinue@genome.tokushima-u.ac.jp)
Mitsuo Itakura (itakura@genome.tokushima-u.ac.jp)

Version: 7 Date: 25 March 2008

Author's response to reviews: see over
Dear the BioMed Central Editorial Production Team:

We sincerely appreciate your e-mail letter of March 19, 2008 concerning formatting changes for our manuscript (MS: 1340804552153919). Please find enclosed our changed manuscript entitled “Lack of association of genetic variation in chromosome region 15q14-22.1 with type 2 diabetes in a Japanese population”. Our responses to three requests are as follows:

1) We formatted our Table 3 without vertical elements.
2) We included the numerical reference in Table 1.
3) We listed all authors’ names in the reference list, if there are 30 or below. We used ‘et al.’ in the reference 3, 4, 5, 6, 13, and 14, because the authors are above 30.

With best regards,

Mitsuo Itakura, M.D., Ph.D.
Professor and Director
Division of Genetic Information
Institute for Genome Research
The University of Tokushima